Author's response to reviews

Title: Percutaneous assisted approach for total hip replacement and its effect on functional rehabilitation compared to the anterolateral approach: study protocol for a randomized controlled trial.

Authors:

Claudia Hendrickx (Claudia.Hendrickx@uantwerpen.be)
Willem De Hertogh (willem.dehertogh@uantwerpen.be)
Ulrike Van Daele (ulrike.vandaele@uantwerpen.be)
Peter Mertens (Peter.Mertens@zna.be)
Gaetane Stassijns (Gaetane.Stassijns@uza.be)

Version: 2  Date: 23 January 2014

Author's response to reviews:

Dear Editor

Following revisions were made according to the remarks by Jessie McGavin:

1. Description of all authors' contributions separately for all authors (UVD was added)

2. A statement of ethical approval was added in the method section of the manuscript.

3. An acknowledgment section was added

We hope our manuscript will meet your standard and are looking forward to your reaction.

On behalf of all authors

Claudia Hendrickx